

Dear editor,

World journal of stomatology.

We are submitting the revised version of our manuscript entitled "ORAL LICHENPLANUS: ETIOLOGY, PATHOGENESIS, DIAGNOSIS, AND MANAGEMENT"(ESPS NO-14320 review).

Title: "ORAL LICHENPLANUS: ETIOLOGY, PATHOGENESIS, DIAGNOSIS, AND MANAGEMENT"

Authors: Naresh Gangeshetty, B Praveen Kumar

Name of journal: world journal of stomatology

ESPS Manuscript NO: 14320

The manuscript has been revised according to the comments made by the reviewers

1. Major revision by adding new references
2. All the comments made by the reviewers are rectified
3. References and typesetting were corrected

please consider the article for publishing.

Thank you sir,

Yours sincerely,

Dr Naresh Gangeshetty,

H No. 11-2-47/6/1, bharatnagar,

siddipet, dist:medak, Telangana, India.